Author's response to reviews

Title: Measures and models for causal inference in cross-sectional studies: arguments for the appropriateness of the prevalence odds ratio and related logistic regression

Authors:

Michael E Reichenheim (michael@ims.uerj.br)
Evandro S F Coutinho (esfcoutinho@ensp.fiocruz.br)

Version: 10 Date: 14 July 2010

Author's response to reviews: see over
Dear Editor(s),

Following instructions, we gave the text a final reading and picked up a handful of very minor slips. Since the manuscript will be eventually displayed for reading, we thought it worth re-submitting the MS once again.

Kind regards,

Michael E. Reichenheim (p/ Evandro S. F. Coutinho)